Author's response to reviews

Title: An investigation of nurses' knowledge, attitudes, and practices regarding disinfection procedures in Italy

Authors:

Alessandra Sessa (ale.sessa@hotmail.it)
Gabriella Di Giuseppe (gabriella.digiuseppe@unina2.it)
Luciana Albano (luciana.albano@unina2.it)
Italo F. Angelillo (italof.angelillo@unina2.it)
and the Collaborative Working Group (CWG) (italof.angelillo@unina2.it)

Version: 5 Date: 21 May 2011

Author's response to reviews:

We have formatted the manuscript as required:

The Author list in the manuscript has been written as they are in the submission system.

The Author groups has been included within the list of authors.
The names of the members of the group has been placed in the Acknowledgements section.
The Author groups contribution as a whole has been included in the authors contributions section.
The Structure of the manuscript has beem corrected.
The reference list has been modified and it now contains up to 30 authors.